Physics 221A Fall 2010 Homework 2 Due Friday, September 10, 2010 at 5pm

Reading Assignment: Sakurai, pp. 23–36, Notes 1, pp. 13 to the end, Notes 2, and Notes 3, pp. 1–11.

Note: Some students asked me if I could post notes before lecture. I will if I can, but I usually revise the notes after my lectures. So I will say “incomplete” next to the set of posted notes if they are likely to change in the near future; in that case, to save paper, you might refrain from printing them out. When I have finished with revisions, I will say “complete” after the notes. Then it should be safe to print them (if you want a hard copy).

Please do Problems 1.5, 2.2, 2.4, and 3.1.